Author's response to reviews

Title: Diabetes and tuberculosis: the impact of the diabetes epidemic on tuberculosis incidence

Authors:

Catherine R Stevenson (n.c.unwin@ncl.ac.uk)
Nita G Forouhi (nf250@medschl.cam.ac.uk)
Gojka Roglic (roglicg@who.int)
Brian G Williams (williamsbg@who.int)
Jeremy A Lauer (lauerj@who.int)
Christopher Dye (dyec@who.int)
Nigel Unwin (n.c.unwin@ncl.ac.uk)

Version: 3 Date: 22 August 2007

Author's response to reviews:

I am responding to the request for formatting changes. I have changed the format of the authors' names to be the same as that in the web based submission; incorporated figure 1 into the text (so that there is no longer a figure); and added an apostrophe to "authors contributions". In addition to these requested changes I have been through the article carefully and corrected some small errors in the references. I have also added one reference (now number 29). This is to provide reference to the source document, rather than simply to a journal article, for the International Standards for TB Care.